Author's response to reviews

Title: Recent alcohol consumption and risk of incident ovarian carcinoma: a pooled analysis of 5,342 cases and 10,358 controls from the Ovarian Cancer Association Consortium

Authors:

Linda E Kelemen (lkelemen@post.harvard.edu)
Elisa V Bandera (banderel@umdnj.edu)
Kathryn L Terry (KTERRY@PARTNERS.ORG)
Mary Anne Rossing (mrossing@fhcrc.org)
Louise A Brinton (brintonl@exchange.nih.gov)
Jennifer A Doherty (jdoherty@fhcrc.org)
Roberta B Ness (Roberta.B.Ness@uth.tmc.edu)
Susanne Krüger Kjær (susanne@cancer.dk)
Jenny Chang-Claude (j.chang-claude@dkfz-heidelberg.de)
Martin Köbel (Martin.Kobel@cls.ab.ca)
Galina Lurie (GLurie@crch.hawaii.edu)
Pamela J Thomson (Pam@cc.hawaii.edu)
Michael E Carney (mcarney@hawaii.edu)
Kirsten Moysich (kirsten.moysich@roswellpark.org)
Robert P Edwards (redwards@mail.magee.edu)
Clareann H Bunker (bunkerc@pitt.edu)
Allan Jensen (allan@cancer.dk)
Estrid Høgdall (estrid@cancer.dk)
Daniel W Cramer (DCRAMER@partners.org)
Allison F Vitonis (avitonis@partners.org)
Sara H Olson (olson@MSKCC.ORG)
Melony King (williamg@umdnj.edu)
Urmila Chandran (chandurr@umdnj.edu)
Jolanta Lissowska (jissowsi@ciw.waw.pl)
Montserrat Garcia-Closas (Montserrat.GarciaClosas@icr.ac.uk)
Hannah Yang (yanghan@mail.nih.gov)
Penelope M Webb (penny.webb@qimr.edu.au)
Joellen M Schildkraut (joellen.schildkraut@dm.duke.edu)
Marc T Goodman (Marc.Goodman@cshs.org)
Harvey A Risch (harvey.risch@yale.edu)

Version: 5 Date: 17 January 2013

Author's response to reviews: see over
Dr Christina Chap, Executive Editor

*BMC Cancer*

Jan 17, 2013

Dear Dr Chap,

We are delighted that our manuscript “Recent alcohol consumption and risk of incident ovarian carcinoma: a pooled analysis of 5,342 cases and 10,358 controls from the Ovarian Cancer Association Consortium” is accepted provisionally for publication as a Research Article in BMC Cancer.

Please find attached version 4 of our manuscript with the following changes:

1. Inclusion of a competing interest statement between the conclusions and authors’ contributions sections
2. Inclusion of authors’ contributions before the acknowledgements section
3. Inclusion of a ‘Methods’ and ‘Results’ heading in the abstract.

We appreciate the opportunity to submit our work to BMC Cancer.

With Warmest Regards,

Linda E Kelemen, MSc, ScD